Author's response to reviews

Title: What training is needed to study the causes of health inequalities in my country? Perspectives from health and demographic researchers in eight African and Asian countries

Authors:

Joke A Haafkens (j.a.haafkens@uva.nl)
Yulia Blomstedt (yulia.blomstedt@epiph.umu.se)
Malin Eriksson (malin.eksson@epiph.umu.se)
Heiko Becher (Heiko.Becher@urz.uni-heidelberg.de)
Heribert Ramroth (Heribert.Ramroth@uni-heidelberg.de)
John Kinsman (john.kinsman@epiph.umu.se)

Version: 2 Date: 20 March 2014

Author's response to reviews: see over
Dear Editor,

Enclosed you find the above manuscript of our paper in which we made the changes you requested:

1.) The email address of all authors is provided on the Title page.
2.) A section on ethics is added to the methods section.
3.) Line Numbering has been inserted
4.) A section on competing Interests has been added to the manuscript

Kind regards,

Joke Haafkens